Author’s response to reviews

Title: Institutionalizing evidence-based practice: an organizational case study using a model of strategic change

Authors:

   Cheryl B Stetler (cheryl.stetler@comcast.net)
   Judith A Ritchie (judith.ritchie@muhc.mcgill.ca)
   Jo Rycroft-Malone (hss401@bangor.ac.uk)
   Alyce A Schultz (alyceme@cox.net)
   Martin P Charns (mcharns@bu.edu)

Version: 6 Date: 19 November 2009

Author’s response to reviews: see over
TO: BioMed Central

RE: Implementation Science Manuscript

Authors: Cheryl B Stetler, Judith A Ritchie, Jo Rycroft-Malone, Alyce A Schultz and Martin P Charns
Title: Institutionalizing evidence-based practice: an organizational case study using a model of strategic change
Journal: Implementation Science
MS: 7199949402290151

DATE: November 19, 2009

This letter is to affirm the fact that we have made the required changes, both major and minor:

<p>| | |</p>
<table>
<thead>
<tr>
<th></th>
<th></th>
</tr>
</thead>
<tbody>
<tr>
<td>1. Authors' contributions: Please indicate that all authors read and approved the final manuscript.</td>
<td>Inserted: “All of the authors take responsibility for the findings reported in this work and have read and approved the final manuscript.”</td>
</tr>
<tr>
<td>2. Table 7: ...suggest incorporating this into the body of the manuscript</td>
<td>Done: Table deleted and content inserted on pages 28-29 as bullets.</td>
</tr>
<tr>
<td>3. Table 5: ...unable to display shading and colours</td>
<td>Shading and colours deleted; spacing adjusted as appropriate to divide the text.</td>
</tr>
<tr>
<td>4. References: ...cite in the text using consecutive numbers</td>
<td>On pg.24, a correction has been made to a citation [one was erroneously cited]; and all numbering is now in the correct order throughout, including tables.</td>
</tr>
<tr>
<td>5. Title: Sentence case</td>
<td>“Institutionalizing evidence-based practice: an organizational case study using a model of strategic change”</td>
</tr>
<tr>
<td></td>
<td></td>
</tr>
<tr>
<td>---</td>
<td>---</td>
</tr>
<tr>
<td>6. Affiliations: …include country in affiliation details</td>
<td>Done for all authors</td>
</tr>
<tr>
<td>7. Tables: …remove visible vertical lines</td>
<td>Done for all tables</td>
</tr>
<tr>
<td>8. Figures: …not include the title …or figure number. The legend and title should be part of the manuscript file after the reference list.</td>
<td>Title and, as needed, legend indicated for Figure 1, Figure 2 and Figure 3 after the reference list and not in the figures.</td>
</tr>
<tr>
<td>9. Figures: … cropped as closely as possible</td>
<td>Done for all three figures</td>
</tr>
<tr>
<td>10. Figures: … in correct orientation</td>
<td>Correct orientation</td>
</tr>
</tbody>
</table>